Author’s response to reviews

Title: Probiotics and infective endocarditis in patients with Hereditary Hemorrhagic Telangiectasia: a clinical case and a review of the literature

Authors:
Evangelo Boumis (eboumis@gmail.com)
Alessandro Capone (alessandro.capone@inmi.it)
Vincenzo Galati (vincenzo.galati@inmi.it)
Carolina Venditti (carolinavenditti@hotmail.it)
Nicola Petrosillo (nicola.petrosillo@inmi.it)

Version: 2 Date: 20 Dec 2017

Author’s response to reviews:

1. Please provide clarification on the consent obtained to publish the Case Report. Did the patient consent to the publication of potentially-identifying information within the case, e.g. age, gender, etc.?

R: The patient signed an institutional form in which he declares that he gives consent to his attending doctors to utilize all clinical data regarding his disease with the purpose of scientific publications, in an anonymous way. We eliminated any reference to age and sex from the document.

2. Please make sure the Abstract and main Manuscript text follows the format outlined in our submission guidelines.

https://bmcinfectdis.biomedcentral.com/submission-guidelines/preparing-your-manuscript/case-report

Please make sure the submission data is also updated with any abstract changes.

R: We changed the structure of the Abstract to follow the format outlined in the submission guidelines (e.g. “Background”, “Case presentation”, “Conclusions”). We updated the changes in the submission data. We controlled and modified the references and modified their format according to submission guidelines.

3. Please provide a list of all the abbreviations used in the manuscript. This list should be placed just before the Declarations section. All abbreviations should still be defined in the text at first use.
R: We added a list of all abbreviations just before the Declarations section. We controlled all the abbreviations in the text to be sure that they are defined in the text at the first use.

4. We noticed the Reviewers suggested Figure 1 to be deleted. If you wish do delete it please do so. If you wish for it to remain within the Manuscript please add a reference to Figure 1 in the text.

R: We deleted figure 1.

5. Please add a “Supplementary files” section where you list the following information about your supplementary material:
   - File name
   - Title of data
   - Description of data

R: We added a “Supplementary files” section with the indication of the name of the file, the title of data and a description of them.

6. Please move Tables 1 and 2 to the end of the manuscript as they are currently too large to include within the manuscript’s text.

R: We moved Table 1 and 2 at the end of the document.

7. Please note that the specific individual contributions of each author needs to be listed in this section individually. Please represent authors' names using their full initials, not their full name, in the Authors’ Contributions section.

R: We listed individual Author’s contribution using their initials.

8. Please remove Supplementary files named ‘lact endoc HHT care checklist.docx’ and ‘response to the comments.docx’ from your Submission as they are no longer needed.

R: We removed supplementary files named “lact endoc HHT care checklist.docx” and “response to the comments.docx”

9. At this stage, please upload your manuscript as a single, final, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.

R: We uploaded a single clean final version of the manuscript.